Author's response to reviews

Title: Consumer input into research: the Australian Cancer Trials website

Authors:

Rachel F Dear (rachel.dear@sydney.edu.au)
Alexandra Barratt (alexandra.barratt@sydney.edu.au)
Sally Crossing (info@cancervoice.org.au)
Phyllis Butow (phyllis.butow@sydney.edu.au)
Susan Hanson (Susan.Hanson@canceraustralia.gov.au)
Martin HN Tattersall (martin.tattersall@sydney.edu.au)

Version: 3 Date: 17 June 2011

Author's response to reviews: see over
Dear Dr Gonzalez Block and Dr Hanney,

RE: MS: 8218973053980245

Thank you for accepting our manuscript “Consumer input into research: the Australian Cancer Trials website.”

We have made the following changes to the format of the manuscript so that the manuscript conforms to the journal style as requested:

- We have edited the author list as requested.
- We have made sure that the Tables have been referenced in the order they appear in the manuscript.
- We have removed the tracking.
- We have formatted the abbreviations into alphabetical order and into one paragraph.
- We have updated the reference list to include up to 30 authors’ names.
- We have removed the visible vertical lines from our tables.
- We have checked the manuscript for typographical errors.

We would appreciate your advice about how best to finalise the layout of Figure 1 (a minor revision requested by your editorial team).

Thank you for your consideration.

Yours sincerely

Rachel Dear MBBS FRACP